Author's response to reviews

Title: Frequency of LCT -13910C>T single nucleotide polymorphism associated with adult-type hypolactasia/lactase persistence among Brazilians of different ethnic groups

Authors:

Rejane Mattar (rejane-mattar@ig.com.br)
Maria S Monteiro (mmont@hcnet.usp.br)
Cibele A Villares (cibele.villares@ig.com.br)
Anibal F Santos (anibal2007@ig.com.br)
Joyce M K Silva (jomatie@gmail.com)
Flair J Carrilho (fjcarril@usp.br)

Version: 4 Date: 29 September 2009

Author's response to reviews:


I had written Nature Genetics instead of Nat Genet. Thus, I corrected in this revised version.